
RHIC Status

• Yellow Cryo is ready
• Yellow Vacuum is ready;
• Yellow PS: should be ready for injection today



RHIC Status

• Blue Established circulating beam, RF captured, Check-out instrumentation ;

• Tried several ramps and found some issues. Fixing now.



Schedule 

Today:

• Open ring for access before 16:00 (CeC, Cryo welder, LeRHIC)

• Fix above outstanding issues and RF cavity issues

• Yellow PS work: polarity check, corrector, …

• More DX training if is possible

• Sweep for over night beam before 16:00

• Yellow injection

• Ramp both blue and yellow 



Where are we

• Plan: Monday OWL has injection setup blue and yellow
Tuesday EVE completes 100 GeV ramp setup (6 only)

• Progress now: Wednesday have could injection setup for blue and yellow
Friday or Saturday could complete 100 GeV ramp setup

Conclusion: we are 2~3 days behind of the best scenario schedule.

First collision with 6x6 for STAR and PHINEX detector check out could be this 
weekend;



Tomorrow Access

1. NO access tomorrow;
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